Author's response to reviews

Title: Jejunal microvilli atrophy and reduced nutrient transport in rats with advanced liver cirrhosis: improvement by Insulin-like Growth Factor I (IGF-I)

Authors:

Prof Inma Castilla de Cortazar (iccottazar@uma.es)  
Maria Pascual (mpascual@ochoa.fib.es)  
Elena Urdaneta (elena.urdaneta@unavarra.es)  
Javier Pardo (jpardo@unav.es)  
Juan Enrique Puche (puche@uma.es)  
Barbara Vivas (berlanga@uma.es)  
Amelia Diaz-Casares (adcasares@iib.uam.es)  
Matias Diaz-Sanchez (mdiaz@unav.es)  
Maria Garcia-Fernandez (igf@uma.es)  
Isabel Varela-Nieto (ivarela@iib.uam.es)  
Alberto Castilla (med003047@saludalia.com)  
Salvador Gonzalez-Baron (sgonzalez@uma.es)

Version: 4  Date: 19 Apr 2004

MS: 1846889770284616

Dear Dr. Christopher Gadd:

Please find enclosed a point-by-point response of referres' comments and a new version of the manuscript including the suggested changes in Abstract and results and a new table (TABLE 1). We have also identified the symbols used in Figure 1 and 2.

We hope that this work would be accepted by BMC Gastroenterology.

Sincerely,

Inma Castilla de Cortazar, MD and PhD  
Prof. of Human Physiology